%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\documentclass[12pt]{article}
\usepackage{amssymb}
\topmargin-2.5cm
\textheight 25cm
\textwidth18.5cm
\oddsidemargin-.7cm
\evensidemargin-.7cm
\pagestyle{empty}
\parindent0cm
\parskip 2.5ex
\newcommand{\re}{\rm Re}
\newcommand{\im}{\rm Im}
\newcommand{\C}{\Bbb C}
\newcommand{\Ds}{\displaystyle}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{document}

\begin{center}
{\bf Math 226.01: Differential Equations and Linear Algebra }
\\
{\bf Quiz 4}
\\
{\bf September 25, 1998}
\end{center}
\vspace*{.4cm}

\underline{{\bf Name:}\hspace*{10cm}}


Find the solution of the following initial value problem
\[
    3y''+14y'-5y=0,\quad y(0)=-4,\quad y'(0)=4.
\]


\vfill
\underline{{\bf Answer:}\hspace*{10cm}}

Sign the pledge:
``On my honor, I have neither given nor received unauthorized aid on this Exam.''

\underline{{\bf Signature:}\mbox{\hspace*{10cm}}}

\end{document}